 Vc/25-Oct-04
Week43 starting 18 oct 04

SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday 18 Oct 04  ~17.30 hrs 

	
	A1
	MBBL3131
	PT 2 HV Cold Test
	

	
	A2
	MBAL3224
	PT 8 Fast De-excitation
	

	
	B1
	SSS053
	Stretched Wire Cold SC21
	

	
	B2
	SSS064
	Prep 5 Pumping + Cool Down
	

	
	C1
	MBAL3234
	PT 6.1 Training 1
	

	
	C2
	MBBL3164
	ICS 4 Final connection
	

	
	D1
	SSS062
	PT 12 Warm Up
	

	
	D2
	SSS066
	Prep 5 Pumping + Cool Down
	

	
	E1
	MBBR1075
	Prep 5 Pumping + Cool Down
	

	
	E2
	MBAL3232
	ICS 2 Connect Magnet (ICS)
	

	
	F1
	MBAL1156
	Prep 3 CDW1 HV
	

	
	F2
	MBAL3227
	ICS 5 Traveller Last
	


10/18/2004 5:44:44 PM    Temperature data are taken from TBSM18 site
Coordinator this week: Nikolay; MTM support on call :  (16-3602).

15 Magnets [ 11 dipoles( 0 with MM)+ 4 SSS +  0 SSSS]   Disconnected after tests last week ( Sat Midnight to Sat midnight)  
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
AOB

RESULTS FROM LAST WEEK
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PROBLEMS OF LAST WEEK

	Date
	Bench
	Problem
	Type
	Application/soft
	Action/advise

	Wednesday _13_M
	TBB1
	IAP @ warm : Trigger on potaim cards
	MTM
	
	One cable was not put back in place after switching from special to arc SSS. Solved by Richard.



	Wednesday _13_M
	TBD2
	High current light flashing
	MTM
	
	Solved by Giorgio, one relay was blocked.

	Wednesday _13_A
	TBD1
	SSW at warm : saturation of the signal in the power supply
	MTM
	
	Nicolay called. Investigation will be pursued tomorrow. Stephane decide to report SSW measurements at warm after the cold test.

	Wednesday _13_A
	TBC1
	Cryo test. 
	ACR
	
	Magnet kept on stand by

	Wednesday _13_A
	TBC1
	Diode test.: first trial: No Vtano, VTcatho signals recorded . 

Test repeated : No data for High Field heaters.
	AB/CO
	lf
	Richard called : cables disconnect by maintenance people. Solved.

Stephane looked for data in lf (transient) but no data: Magnet is protected but discharges data not visible. Stop the test and Richard called. Data stored in lf but in archieved mode: problem with lf DAQ?

Swith off/on the crates, restart DAQ, trigger one HF discharge. It works. Quench repeated.

	Wednesday _13_A
	TBB1
	SSW on TBBI: work station not working : one hard disk crashed (machine 33).
	AB/CO
	
	Richard called Magnus. Solved for this time.

	Thursday_14_M
	TBF1
	IAP @cold: freesing of curremnt leads. Test stop
	ACR
	
	Tprobe faulty. Solved by ACR team

	Thursday_14_A
	TBD1
	Sun Screen not working? 
	MTM
	
	Connector disconnected. Solved

	Thursday_14_A
	TBF1
	Diode test: faulty quench at 150 A on 2100 magnet
	MTM
	
	Checked by Stephane and Richard. Outside the coil, in the bus bar

Repeatedd with reduced ramp rate 20 A/s. OK

	Thursday_14_A
	TBD1
	Problem with powering CD circuit  
	IT
	
	Server problem. Solved.

	Thursday_14_N
	TBD1
	Not possible to power EF circuit 
	MTM

AB/CO
	
	Not correct inductance put in the power seeting.

TEMA hung. Many times.

Then sunday OK but quenching 3 times below nominal

	Friday_15_A
	TBD1
	MQ strength: Process hung “checking motor status”.

Slow PA abort at 11850 A
	MTM

IT
	
	Switch off the motors, reset of the motor controller

	Friday_15_N
	TBA2
	3224 magnet : Faulty quench when PC switched in stand by mode : inductance value not cortrectly selected.
	MTM


	
	

	Saturday_16_A
	SM18
	Oxygen deficiency in SM18 2% below the normal (level was at 19 .1 %): Alarm level 1 (for information)
	MTM


	
	A.Siemko saked to keep the main entrance door opened.

Alarm cleraed 2 h after.

	Saturday_16_N

Sunday _17_M, A
	TBD1
	Wire broken during XY measurements in AP2.

Server crashed.

XY measurements: Values out of tolerances.

Wire not coming in the right position.

Resistance value not correct (too low,0.1 to 0.3 mOhm).
	MTM


	
	Relaunched.

Stephane, Richard called. Advise to change the DB file. No change.

Richard found that wire is touching the anticryostat. Repeated with a smaller path.

Repeated, results good and no problem with resistance but still not coming in the right position.: reason : No good alignment bewteen short anticryostat and magnet.

	Saturday_17_A
	TBB1
	Label of the power cable GH and IJ corcetors missing.
	MTM


	
	A.Siemko asked to keep the main entrance door opened.



	Sunday_17_A
	TBA2
	After quench number 6 faulty quench when we start  the ramp (10 A) switching in stand by mode.
	AB/PO


	
	Richard came and investigated the problem : inside the power supply RMA. He called AB/PO. AB/PO piquet rest the digitla controller.

	Sunday_17_A
	TBB1
	Not possible to power CD circuit. 
	MTM


	
	Richard investigated the problem :with PO team he managed to power the sextupole disabling the quench detection and the feedback loop. 

It was found that correct inductance to be used in 5 mH and not 18 mH as foreseen. 

	Sunday_17_N
	TBC1
	MBAL3234: During first  training quench one heater YT121, jump did not discharged completely in the middle of dischage (250 ms after). Vt saturated at 250 V.
	MTM


	
	Stephane informed. Test stopped.He advise to check:

-Resistance of heaters

-Electrical integrity.










Light green : Connect (up to Cryo pump down)	24 h


Blue :  Wait + Cool down				26 h


Ivory: Tests at Cold					36 h


Red :  Wait + Warm up					12 h


Dark Green: Disconnect					12 h
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